Author’s response to reviews

Title: Relationship between carotid intima-media thickness and coronary angiographic findings: a prospective study

Authors:

Ugur Coskun (ucoskun1@yahoo.com)
Ahmet Yildiz (drayildiz@yahoo.com)
Ozlem Batukan Esen (ozlembat@hotmail.com)
Murat Baskurt (drmuratbaskurt@yahoo.com)
Mehmet A Cakar (makifcakar@yahoo.com)
Kadriye Orta Kilickesmez (kadryeilickesmez@yahoo.com)
Lutfu A Orhan (lutfu.orhan@yahoo.com)
Seyma Yildiz (drseymayildiz@yahoo.com)

Version: 4 Date: 9 December 2009

Author’s response to reviews:

Dear sir,

This is the final version of the manuscript titled ‘Relationship between carotid intima-media thickness and coronary angiographic findings: a prospective study’. We have changed the conclusion into more modest one.

I would like to thank you in advance for your help.

Sincerely,

Ahmet Yildiz M.D
Cardiologist